Author's response to reviews

Title: Successful application of Technetium-99m-labeled octreotide acetate scintigraphy in detection of Ectopic Adrenocorticotropic-Producing Bronchial Carcinoid Lung Tumor

Authors:

Armaghan Fard Esfehani (fardesfahani@sina.tums.ac.ir)
Maryam Chavoshi (chavooshi@sina.tums.ac.ir)
Mohammad Hadi Noorani (nooranimh@sina.tums.ac.ir)
Mohsen Saghari (saghari@sina.tums.ac.ir)
Babak Fallahi (fallahnisichan@tums.ac.ir)
Davood Beiki (beikidav@sina.tums.ac.ir)
Mohammad Eftekhari (eftekharimoh@sina.tums.ac.ir)
Majid Assadi (assadipoya@yahoo.com)

Version: 2 Date: 7 April 2010

Author's response to reviews: see over
Dr. Michael Kidd  
**Editor-in-Chief**  
Journal of Medical Case reports

Thank you for the precise analysis of the manuscript. Revision according to the comments of reviewers was performed; detailed description of changes based on the reviewers points are listed below.

1) **Figure 2:**... well defined mass about 22 mm in the lower lobe of the right lung...  
   Image is flipped vertically (!!!).  
The lesion appears to be in the middle lobe, not in the lower lobe (right lung !!!).  
The arrow does not point to the well defined round mass (which corresponds to the bronchial carcinoid lung tumor), but more ventrally to an ill-defined triangular shaped low attenuated area (that probably represents dystelectasis).

Thanks, the location of lesion was checked again and the presented figure is replaced with high quality picture and the arrow point out correctly.

2) **Quality of written English:** Needs some language corrections before being Published.

Thanks for your attention. The manuscript was checked grammatically.

All authors read and approved the final manuscript. I shall be greatly obliged if you would kindly inform me of your proceedings.

   Sincerely yours,

   Majid Assadi